Author's response to reviews

Title: Investigating preferences for support with life after stroke: a discrete choice experiment.

Authors:

Christopher R Burton (c.burton@bangor.ac.uk)
Emily Fargher (e.fargher@bangor.ac.uk)
Catrin Plumpton (c.o.plumpton@bangor.ac.uk)
Gwerfyl W Roberts (hss006@bangor.ac.uk)
Heledd Owen (heledd.owen@bangor.ac.uk)
Eryl Roberts (eryl.roberts@bangor.ac.uk)

Version: 3 Date: 23 January 2014

Author's response to reviews: see over
22 January 2014

The Editor
BMC Health Services Research

Re: MS: 1931509631978989
Investigating preferences for support with life after stroke: a discrete choice experiment.

We are pleased that the Editors have accepted this manuscript, and have made the following changes to the format of the manuscript as follows:

1. The manuscript now includes a Title Page with the details of authors, and the postal address of the corresponding author, provided.

2. Figure files have been uploaded separately and in the correct order. These files no longer contain information on the Figure number or its title. The legend and titles of Figures have been provided in a list after the references. We have revisited the listing and numbering of all Figures and Tables in the text, ensured these are cited in order, and removed the ‘Insert Table’ instructions from the text.

Please do not hesitate to contact us if any further information is required.

Yours sincerely

Dr Christopher R Burton DPhil BN RGN
Senior Research Fellow
College of Health and Behavioural Sciences, Bangor University